Author's response to reviews

Title: Nonsteroidal anti-inflammatory drug use and Alzheimer’s disease risk: the MIRAGE Study

Authors:

Agustin G Yip (agyip@bu.edu)
Robert C Green (rcgreen@bu.edu)
Matthew Huyck (mhuyck@bu.edu)
L Adrienne Cupples (adrienne@bu.edu)
Lindsay A Farrer (farrer@bu.edu)

Version: 3 Date: 14 August 2004

Author's response to reviews:

As per Dr Lindsay's comments we have revised the abstract as follows:

(1) We sought TO examine...
(2) ...ALTHOUGH THE INTERACTION TERM WAS NOT STATISTICALLY SIGNIFICANT...